Disagreement over the persistence of earnings components: evidence on the properties of management-specific adjustments to GAAP earnings

Young-Soo Choi · Stephen Lin · Martin Walker · Steven Young

Published online: 7 July 2007 © Springer Science+Business Media, LLC 2007

Abstract We examine disagreement between management and Thomson Datastream over the persistence of earnings components. Using income statement and footnote disclosures, we identify the source and properties of disputed items. Disagreements typically reflect opaque reporting practices (for example, in the case of transitory operating items) and restrictive classification rules (for example, in the case of discontinued operations). Incremental and relative value relevance tests suggest that the majority of management-specific adjustments reflect appropriate classification of earnings components by insiders. Nevertheless, evidence consistent with strategic disclosure does emerge for a subset of management adjustments.

Keywords Non-GAAP earnings · Transitory items · Disagreement

JEL Classification M4

S. Young e-mail: s.young@lancaster.ac.uk

S. Lin

M. Walker Manchester Accounting and Finance Group, Manchester Business School, University of Manchester, Manchester M13 9PL, UK e-mail: Martin.Walker@mbs.ac.uk

Y.-S. Choi (⊠) · S. Young Department of Accounting & Finance, Lancaster University Management School, Lancaster University, Lancaster LA1 4YX, UK e-mail: choiy@exchange.lancs.ac.uk

School of Accounting, College of Business Administration, Florida International University, 11200 SW 8th Street, University Park, Miami, FL 33199, USA e-mail: lins@fiu.edu

1 Introduction

Interest in non-GAAP earnings reporting increased dramatically following the disclosure of supplementary earnings metrics by some U.S. firms in their quarterly press releases (Bradshaw and Sloan 2002; Doyle et al. 2003; Bhattacharya et al. 2003; Lougee and Marquardt 2004; Marques 2006).¹ Despite concern over the quality of such metrics (Turner 2000; Business Week 2001), studies using hand-collected data reveal that non-GAAP earnings disclosures often conform precisely to sustainable earnings proxies derived by analysts and other sophisticated financial statement users, suggesting that non-recurring earnings components are often transparently transitory.² If non-GAAP earnings merely adjust for transitory items that are transparently observable to financial statement users then the costs and benefits of such disclosures are unclear.

Differences between the adjustments made by management and analysts do occur in a significant fraction of cases, however (Bhattacharya et al. 2003; Marques 2006). Yet few studies have explicitly analyzed the properties of earnings components over which management and analysts disagree. A notable exception is Marques (2006), who decomposes the difference between GAAP earnings and non-GAAP earnings into two parts: the disparity between GAAP earnings and IBES actual earnings and any incremental difference attributed to management. Results reveal a stark difference between the market's assessment of adjustments made by IBES and the incremental adjustments made by management. Whereas investors view analysts' adjustments as an appropriate elimination of transitory items, they view incremental adjustments attributable to management as an inappropriate exclusion of recurring items. These findings support Lougee and Marguardt's (2004) evidence that strategic disclosure aimed at presenting firm performance in a favorable light represents an important determinant of non-GAAP earnings reporting. The alternative view is that such metrics more accurately reflect insiders' private information regarding the firm's true earnings power (Bhattacharya et al. 2003).

Our paper aims to shed new light on non-GAAP earnings reporting. Like Marques (2006) we focus on discrepancies between management and analysts over the classification of earnings components. However, our analysis differs from Marques' in several respects. Most notably, whereas Marques (2006) analyses the aggregate difference between non-GAAP earnings and IBES actual earnings, we identify the source of disagreement and examine the properties of each disputed component separately. Our disaggregated approach reflects the fact that discrepancies between management and analysts can arise for two distinct reasons: management may exclude additional earnings components beyond those omitted by analysts; or

² Bhattacharya et al. (2003) report that non-GAAP earnings correspond to IBES actual earnings in approximately 65% of cases. Such is the level of correspondence that several studies use IBES actuals as a proxy for management-reported non-GAAP earnings (Brown and Sivakumar 2003; Doyle et al. 2003; Doyle and Soliman 2004; Landsman et al. 2006).



¹ We use the nomenclature "non-GAAP" earnings (or EPS) throughout the paper when referring to supplementary earnings metrics disclosed by management. While prior U.S. research often labels such disclosures "pro forma earnings", our tests relate to the U.K. where the term pro forma earnings is not used.

analysts may exclude supplementary items retained by management. Pooling both types of exclusions into an aggregate measure of disagreement can lead to ambiguous conclusions regarding managements' reporting incentives. For example, evidence that aggregate disagreements are value relevant implies inappropriate exclusion of recurring items by management if incremental exclusions are unique to insiders. On the other hand, if exclusions are unique to analysts, then results are more consistent with management correctly identifying and *including* recurring items that analysts inappropriately classify as transitory. The distinction is important because prior research provides evidence consistent with both scenarios. For example, Matsumoto (2002), Burgstahler and Eames (2003) and Doyle and Soliman (2004) document how management appears to use non-GAAP earnings to strategically meet and manage analysts' earnings per share (EPS) forecasts. Meanwhile, Doyle et al. (2003) and Landsman et al. (2006) show that analysts' exclusions (GAAP earnings minus IBES actual earnings) are forecast and value relevant, suggesting that analysts incorrectly exclude persistent earnings components.

Using a sample of non-GAAP EPS disclosures made by U.K. firms from 1993 to 2001, we identify disagreement between management and Thomson Datastream over the treatment of earnings components. We focus on Thomson's recurring earnings metric rather than on IBES actual earnings because the latter does not facilitate item-byitem reconciliation to GAAP earnings whereas the Thomson number does. For the 45% of sample cases where management and Thomson differ in their adjustments to GAAP earnings, we use income statement and footnote disclosures to manually reconcile adjustments made by each party. "Management exclusions" are defined as items excluded from earnings by management but not by Thomson. Conversely, "management inclusions" represent reversals by management of items excluded by Thomson. Our approach combines features of Gu and Chen (2004), who examine the properties of non-recurring earnings components included and excluded by First Call analysts, and Brown and Sivakumar (2003), who compare operating earnings reported by managers and analysts with a recurring earnings proxy derived by Standard and Poor's.

Descriptive statistics reveal that although non-operating items represent a large fraction of total exclusions in our sample, they account for a disproportionately small fraction of disputed cases. The underlying nature of non-operating gains and losses supplemented by clear income statement disclosure appears to render such items transparently transitory, thereby reducing the scope for disagreement. Instead, disagreements center on operating items and discontinued operations where the idiosyncratic nature of gains and losses combined with opaque disclosure create confusion over the persistence of certain items.³ Decomposing net disagreements, we find that managers reverse Thomson exclusions almost twice as often as they exclude additional items beyond those omitted by Thomson, and the majority (71%) of these reversals involve management adding back losses. These findings are inconsistent with claims that management-specific adjustments to GAAP earnings typically reflect opportunism aimed at presenting firm performance in a more favorable light.

³ Doyle et al. (2003) present evidence that analysts' classification errors are largely confined to operating items. In particular, they partition the difference between GAAP EPS and IBES actual EPS into special (non-operating) items and other (operating) exclusions and find that recurring exclusions are confined to the latter category where financial statement disclosures are especially opaque (Doyle et al. 2003: 151).



Our analysis of the incremental value and forecasting relevance of managementspecific adjustments suggests that disagreement typically reflects managers' superior ability to correctly classify earnings components. Aggregate management exclusions are value and forecast irrelevant, indicating that managers identify additional transitory earnings components overlooked by Thomson. In contrast, management inclusions are incrementally value and forecast relevant, suggesting that managers correctly identify and retain recurring earnings components misclassified by Thomson as transitory. A broadly similar story emerges when we decompose adjustments into net gains and losses. Excluded gains and losses are both value irrelevant whereas included losses are highly value relevant. Evidence consistent with strategic disclosure does emerge in relation to included gains, however, which on average are found to be value irrelevant. Such cases represent less than 10% of the non-GAAP disclosure sample and approximately 20% of disputed adjustment cases. Therefore while the evidence suggests that some firms may use non-GAAP earnings disclosures to artificially boost reported performance, opportunism does not appear to characterize our data on average.

Our analysis offers novel insights relevant to the growing body of research on non-GAAP earnings reporting. Most notably, we investigate the specific source(s) of disagreement between management and outsiders over the properties of earnings components. The benefits of this approach are twofold. First, we can implement tests capable of clearly distinguishing between adjustments driven by opportunism and those that reflect managers' superior information about the persistence of earnings components. Contrary to Marques (2006), we find management-specific adjustments often reflect appropriate elimination (inclusion) of transitory (permanent) items. Second, we can provide evidence on the sources of disagreement. Our findings suggest that disagreements often reflect opaque and ambiguous reporting practices such as inadequate disclosure of information on non-recurring operating items and a restrictive definition of discontinued operations. A further distinguishing feature of our analysis is its U.K. focus. Regulations governing non-GAAP earnings reporting in the U.K. differ from those in the U.S. in several important respects. In particular, Financial Reporting Standard No. 3: Reporting Financial Performance (FRS 3) permits U.K. firms to report supplementary EPS numbers on the face of the income statement alongside GAAP EPS. To ensure transparency, however, firms are required to reconcile non-GAAP EPS to the corresponding GAAP number by way of a note to the income statement. It is these disclosures (which until recently were unavailable in the United States) that facilitate our disaggregated analysis of disputed earnings components. In addition, while FRS 3 seeks to improve the transparency with which non-recurring earnings components are reported, several factors combine to restrict disclosure clarity and consistency (Davies et al. 1999). Scope for confusion over the persistence of certain items makes the United Kingdom an attractive experimental setting in which to explore the properties of disputed adjustments to GAAP earnings.

The remainder of paper proceeds as follows. The next section reviews the origins and governance of non-GAAP earnings in the U.K., paying particular attention to the provisions of FRS 3. Section 3 describes our sample and data. Section 4 reviews the method we use to decompose disagreement, and Section 5 presents evidence on the informativeness of disputed earnings components. Section 6 concludes.



2 Earnings reporting in the U.K.

Before the introduction of FRS 3 in June 1993, GAAP EPS in the U.K. was viewed as the primary statistic for business valuation and performance measurement (e.g., Smith 1992: 63). According to the Accounting Standards Board (1992), this led to naïve interpretations of financial performance by investors and widespread earnings manipulation by management. The Board responded by issuing FRS 3, a key theme of which is that the concept of sustainable earnings is too subjective to be calculated on a standardized basis. FRS 3 therefore shifts attention away from a mechanistic pre-occupation with a single earnings number by de-emphasizing GAAP EPS. This is achieved by calculating EPS *after* charging all transitory items and discontinued operations. The resulting all-inclusive definition of earnings ensures that in most cases the figure merely serves as starting point for further analysis. Not surprisingly, Lin and Walker (2000) find that GAAP EPS computed under FRS 3 is significantly less informative than a proxy for permanent earnings.

In recognition of the potential problems of using GAAP EPS for valuation and performance measurement, FRS 3 allows management the option of reporting additional non-GAAP EPS metrics on the face of the income statement alongside the GAAP number. Non-GAAP disclosures give management the opportunity to report a customized view of performance and highlight important firm-specific earnings streams. These disclosures form part of the audited financial statements, and to maintain transparency a full reconciliation to GAAP EPS is required.

In addition to changing how EPS is defined and reported, FRS 3 also restructured the income statement with the aim of improving the clarity with which earnings components are reported. Key innovations included disaggregating results into continuing operations, acquisitions, and discontinued operations and separate disclosure (below operating profit and interest) of three categories of non-operating item: gains and losses on disposals of operations, costs of fundamental reorganization and restructuring, and gains and losses on the sale of fixed assets.⁴

While these structural changes are designed to improve transparency, considerable room for confusion remains. First, disclosure rules conflate certain recurring and non-recurring activities. For example, discontinued operations under FRS 3 are exactly that: operations discontinued (sold or ceased permanently) in the financial year or shortly after the year-end.⁵ Operations failing to satisfy this condition are automatically classified as part of continuing operations in spite of their nonrecurring nature. Similarly, *all* gains and losses on asset disposals are automatically excluded from operating profit following the requirement to report such transactions in a separate category below operating profit. Firms engaged in fixed asset trading (for example, bars in the case of a brewer or oil acreage in the case of an oil company) are therefore prohibited from including a potentially important stream of recurring earnings in operating profits. Second, management possesses significant

⁵ The U.K. approach contrasts with APB 30 in the United States and International Accounting Standard No. 35, both of which allow operations to be classified as discontinued as soon as a detailed formal plan for disposal has been adopted and announced or when the enterprise has already contracted for the disposal.



⁴ See Lin and Walker (2000) and Walker and Louvari (2003) for further details of FRS 3.

discretion over the positioning of earnings components. For example, although FRS 3 requires fundamental restructuring and reorganization costs be disclosed below the operating profit line, it provides no guidance on distinguishing fundamental activities from non-fundamental ones. Consequently, whether restructuring and reorganization charges are disclosed as part of operating profit or reported separately below operating profit is a highly subjective exercise that depends on management's judgment (Davies et al. 1999: 1513). More generally, since neither FRS 3 nor U.K. Company Law formally define "operating profit" the decision on whether to exclude a transaction from operating profit depends on management's definition of operating activities. Discretion over the classification of non-recurring items as either operating or non-operating is important because it partly determines the transparency with which they are reported. In particular, whereas key nonoperating items are separately and transparently disclosed on the face of the income statement, no standardized reporting method exists for transitory operating components. The result is often inconsistent and opaque disclosure of these items (Davies et al. 1999: 1513). Non-GAAP EPS disclosures provide management with a means of cutting through this confusion by stripping out ambiguous earnings components. Critics, however, argue that managers use such disclosures to further obscure the distinction between transitory and recurring earnings components.

Walker and Louvari (2003) provide preliminary evidence on the factors motivating U.K. firms to disclose non-GAAP EPS. They conclude that the probability of non-GAAP EPS disclosure is driven by two factors: the general disclosure attitude of the firm (that is, transparent versus opaque) and the desire to present a more favorable earnings profile (i.e., transforming GAAP losses in non-GAAP profits). While Walker and Louvari's (2003) evidence suggests that opportunistic motives may drive non-GAAP EPS reporting in the U.K., they do not examine the properties of excluded items or how these adjustments correspond to those made by other informed parties.

3 Sample, data, and descriptive statistics

3.1 Sample selection

We hand-collect non-GAAP EPS disclosures from firms' published financial statements at three time points: calendar years 2001 and 1996, plus firms' first financial statements published after the introduction of FRS 3 in June 1993. The sample frame at each time point comprises the 500 largest London Stock Exchange-listed non-financial firms ranked by market capitalization. Firms domiciled outside the U.K., firms whose primary results are reported in a foreign currency, and firms without earnings data on Thomson Datastream are excluded (with replacement). Some of our tests also use IBES data, resulting in the loss of 199 observations. Details of our sample selection procedure are reported in panel A of Table 1. The proportion of firms reporting non-GAAP EPS increased monotonically during the sample period. By 2001 three-quarters of sample firms disclosed non-GAAP EPS compared with 40% in 1993/94. Whereas non-GAAP earnings reporting in the



United States is largely confined to high-tech industries (Bhattacharya et al. 2003; Johnson and Schwartz 2005; Lougee and Marquardt 2004), non-GAAP reporters in the United Kingdom are drawn from a broad cross-section of industries. Untabulated results reveal 28 (out of 30) Datastream level-4 non-financial industry groups are represented in the 2001 discloser group, with no single industry accounting for more than 11% of the sample.

Items excluded from GAAP EPS by management are summarized in panel B. Non-operating items (in particular gains and losses on sale or termination of operations, gains and losses on fixed asset disposals, and fundamental restructuring and reorganization costs) account for a large fraction of excluded items. Management also excludes a range of operating items including asset impairments, merger and acquisition costs, and other items (both specified and unspecified).⁶ The dramatic rise in the number of firms adding back goodwill amortization in 2001 reflects the introduction of FRS 10: *Goodwill and Intangible Assets* in December 1998 and its requirement to capitalize and amortize purchased goodwill. While many adjustment categories in panel B are clearly transitory others are less obviously so.

3.2 Non-GAAP earnings and the presence of transitory earnings components

As a precursor to our analysis of disagreement, this section explores why U.K. firms report non-GAAP EPS and presents preliminary evidence on the characteristics of managements' adjustments. FRS 3 allows non-GAAP EPS reporting to help management isolate persistent earnings components. We therefore expect the propensity for non-GAAP EPS disclosure to be higher when GAAP earnings contain a significant transitory element. Table 2 presents evidence on the properties of GAAP EPS relative to two recurring earnings proxies, conditional on management's decision to disclose non-GAAP EPS. Our first recurring earnings proxy is actual EPS reported by IBES (Bradshaw and Sloan 2002; Brown and Sivakumar 2003; Doyle et al. 2003; Bhattacharya et al. 2003; Landsman et al. 2006; Marques 2006). Our second proxy is Thomson Datastream EPS before all non-recurring items. Similar to the procedure used by Standard and Poor's and documented by Brown and Sivakumar (2003), Thomson analysts use financial statement disclosures to identify non-recurring earnings components.⁷ We define

⁷ Although Thomson provides few details of the specific process used to identify non-recurring earnings components, comparison of Thomson's non-recurring items with those classified as "exceptional" in firms' financial statements reveals material differences. Thomson therefore appears to take its own view on the persistence of earnings components rather than relying exclusively on classifications provided by management. Further, total non-recurring items identified by Thomson include transactions reported by management within both non-operating and operating profit, indicating that Thomson's adjustment process recognizes the discretion afforded to U.K. management over the positioning of transitory items. Thomson scrapped its Datastream Company Accounts service in April 2004 and as a result no longer reports this definition of non-recurring items.



⁶ Other specified operating items comprise a broad class of components including fixed asset and goodwill impairments, costs associated with acquisitions and demergers, aborted acquisition costs, compensation for loss of office, costs associated with integration, reorganization, restructuring, rationalization and redundancy, provisions against loans to associate undertakings, pension scheme credits, exploration costs written off, and foreign currency gains.

	Sample	years ^a		Pooled sample
	1994	1996	2001	
Firms disclosing non-GAAP EPS	195	264	354	813
Firms not disclosing non-GAAP EPS	305	236	146	687
	500	500	500	1500
Less:				
Firms with missing IBES actual EPS or price data	(39)	(27)	(133)	(199)
Final sample	461	473	367	1301
Sample firms partitioned by disclosure type				
Firms disclosing non-GAAP EPS	181	253	278	712
Firms not disclosing non-GAAP EPS	280	220	89	589
	461	473	367	1301

Table 1 Sample selection method and characteristics of earnings components excluded by management

Panel B: classification of excluded items

	% of firr	ns disclosi	ng non-GA	AP EPS ^b
Excluded item category	1994	1996	2001	Pooled
Below operating profit (i.e., non-operating activities)				
Profit (loss) on sale or termination of operations	47.0	48.6	45.7	47.1
Profit (loss) on sale of fixed assets or investments	34.8	29.6	27.7	30.2
Reorganization, restructuring and integration costs	19.3	29.6	41.7	31.7
Other special gains (losses)	10.5	11.9	15.1	12.8
Other non-operating adjustments	2.8	6.3	10.4	7.0
Exceptional dividend, interest or taxation	3.9	4.3	1.4	3.1
Within operating profit (i.e., operating activities)				
Amortization of goodwill	0.6	0.0	74.1	29.1
Impairment, diminution or write-off of goodwill	0.6	0.0	9.4	3.8
Revaluation or impairment of fixed assets	8.8	5.5	13.7	9.6
Acquisition and merger/demerger costs	1.1	3.6	5.8	3.8
Other specified exceptional costs	9.4	14.6	22.3	16.3
Other unspecified exceptional items	5.0	3.2	7.2	5.2
Other				
Discontinued operations	5.5	4.3	3.6	4.4
Provisions	2.2	2.4	6.1	3.8
Adjustment of weighted no. of shares	0.6	1.6	0.4	0.8

^a In each sample year, all U.K.-domiciled non-financial firms in the Datastream active and inactive files with earnings data reported in pounds sterling are ranked by year-end market capitalization and the top 500 selected

^b Column totals exceed 100% because management exclude multiple items from GAAP earnings



Panel A: descriptive statistics	iptive statistics					
2		EPS metric ^a			p-value for difference ^b	
		GAAP	TD	IBES	GAAP vs. TD	GAAP vs. IBES
Non-disclosers $(N = 589)$	(N = 589)					
Mean		19.79	20.67	22.37	0.230	0.086
Std		31.33	23.61	39.93		
Median		16.80	16.46	17.00	0.002	0.142
Disclosers $(N = 712)$	= 712)					
Mean		13.94	21.43	21.64	0.001	0.001
Std		31.14	19.97	20.45		
Median		14.63	18.21	18.89	0.001	0.001
Panel B: value relevance	relevance					
Model: ^c $P_t = \alpha_t$	$\alpha_0 + \alpha_1 BVPS_t + \alpha_2 EPS_t^k + \varepsilon_t$	E _r				
EPS metric	\$	α1	α_2	Adj-R ²	Vuong test: ^d	
					Z-stat	p-value
Non-disclosers $(N = 589)$	(N = 589)					
GAAP	227.61	0.432	3.320	0.431	I	I
TD	184.94	0.340	6.247	0.486	-1.33	0.183
IBES	240.31	0.490	1.779	0.395	0.91	0.363
Di	= 712)					
GAAP	226.15	0.490	3.027	0.378	I	I
TD	126.24	0.260	8.631	0.586	-4.57	0.001
IBES	131.80	0.787	8 004	0 565	-350	0.001

Part C: predictability Model: $CCF_{i,1} = \beta_0 + \beta_1 EPS_i^* + \eta_i$ EPS metric β_0 β_1 Adj. R^2 Vuong test ^d PNOn-disclorers ($N = 564$) β_0 β_1 Adj. R^2 Vuong test ^d EPS metric β_0 β_1 Adj. R^2 Vuong test ^d PNOn-disclorers ($N = 564$) $2.3.64$ 1.022 0.383 -2.40 0.016 Disclorers ($N = 680$) 3.56 1.092 0.383 -2.40 0.026 Disclorers ($N = 680$) 3.767 0.772 0.206 1.14 0.233 Disclorers ($N = 680$) 3.767 0.772 0.270 -2.40 0.016 Disclorers ($N = 680$) 3.767 0.772 0.270 -2.40 0.020 Disclorers ($N = 680$) 3.767 0.772 0.270 -2.40 0.020 Disclorers ($N = 680$) 3.767 0.772 0.270 -2.40 0.020 Disclorers ($N = 680$) 0.712 0.270 -2.40 0.020 -2.40 0.020 Disclorers ($N = 680$	L	Table 2 continued					
••.] •[]		² anel C: predictability					
EPS metric β_0 β_1 $Adj\cdot R^2$ Vuong test ⁴ Z-stat p -value Z^{-stat} p -value Z^{-stat} Z^{-stat} p -value Z^{-stat} Z^{-stat} p -value Z^{-stat} Z^{-stat} p -value Z^{-stat} Z^{-stat} Z^{-stat} p -value Z^{-stat} Z^{-stat		$\text{Model:}^{\circ} OCF_{i+1} = \beta_0 + \beta_1 EPS_i^k + \eta_i$					
Non-disclosers ($V = 564$) GAP 23.64 1.022 0.383 – $-$ 240 0.016 DAP 23.64 1.022 0.383 – $-$ 240 0.016 DES 30.70 0.584 0.206 1.14 0.253 DES 30.70 0.584 0.206 1.14 0.253 DES 30.70 0.584 0.206 1.14 0.253 DES 3.0.70 0.572 0.200 – $-$ 240 0.001 DES 1.139 1.717 0.650 – -9.09 0.001 DES 1.139 1.717 0.650 – -4.40 0.001 DES 1.149 0.201 $-$ 2.40 0.001 DES 1.159 1.717 0.650 – -4.40 0.001 DES 1.152 0.679 -9.09 0.001 DES 1.152 DES measured on an undusted bars. 0 Probability values for tests of paired differences in means and medians 0 Variable definitions are as follows: P is stock price measured 3 months after the fixed year-end; BYS is book value of shareholders' funds per share; EPS^{4} is earmings per share computed for metric k ($k = GAAP$, TD, and DES PS); and OCF is operating cash flow per share.	Ž	3PS metric	β_0	β_1	Adj-R ²	Vuong test ^d	
Non-disclosers (V = 564) Non-disclosers (V = 564) 23.64 1.022 0.383 - - - TD 8.65 1.699 0.599 -2.40 0.016 BES 30.70 0.584 0.206 1.14 0.253 Disclosers (V = 680) 37.67 0.772 0.270 - - Disclosers (V = 680) 37.67 0.772 0.270 - - - Disclosers (V = 680) 37.67 0.772 0.270 - - - - Disclosers (V = 680) 37.67 0.772 0.270 - - - - - - - - - - - - - - 0.001 -	j						<i>p</i> -value
GAP 23.64 1.022 0.383 - - - TD 8.65 1.699 0.599 -2.40 0.016 BES 30.70 0.584 0.206 1.14 0.253 Disclosers (N = 680) 37.67 0.772 0.270 - - - Disclosers (N = 680) 37.67 0.772 0.270 - - - - - - - - - - - - - - 0.253 0.270 - - - 0.233 0.201 0.270 - - - - - - - 0.233 0.001 0.203 0.114 0.233 0.270 - - 0.231 0.270 - - 0.201 0.201 0.201 0.201 0.201 0.201 0.201 0.201 0.201 0.201 0.201 0.201 0.201 0.201 0.201 0.201 0.201 0.201 0.201		Non-disclosers ($N = 564$)					
TD 8.65 1.699 0.599 -2.40 0.016 BES 30.70 0.584 0.206 1.14 0.253 Disclosers (N = 680) 37.67 0.772 0.270 - - - TD 37.67 0.772 0.270 - - - - - - - - - - 0.001 11.4 0.253 1.14 0.253 1.14 0.253 1.14 0.253 1.14 0.253 1.14 0.253 1.253 1.82 0.270 - - - - - - 0.001 1.14 0.201 1.14 0.203 1.011 1.14 0.203 1.011 1.14 0.201 1.14 0.201 1.11 1.14 0.201 1.14 0.201 1.11 1.11 1.11 1.11 1.11 1.11 1.11 1.11 1.11 1.11 1.11 1.11 1.11 1.11 1.11 1.11 1.11 <t< td=""><th></th><td>GAAP</td><td>23.64</td><td>1.022</td><td>0.383</td><td></td><td>1</td></t<>		GAAP	23.64	1.022	0.383		1
IBES30.700.5840.2061.140.253Disclosers (N = 680)37.670.5720.570TD37.670.7720.270TD8.231.8820.679-9.090.001DISC8.231.7170.620-4.400.001DISC8.231.7170.620-4.400.001DISC8.231.7170.620-4.400.001BES9.1591.7170.620-4.400.001BES is actual EPS reported by IBES. All EPS numbers refer to basic EPS measured on an unadjusted basis0.650-4.400.001Debability values for tests of paired differences in means and medians0.650-4.400.001Debability values for tests of paired differences in means and medians0.650-4.400.001C Variable definitions are as follows: P is stock price measured on an unadjusted basis0.650-4.400.001Debability values for tests of paired differences in means and medians0.650-4.400.601-6.009D Probability values for tests of paired differences in means and medians0.650-4.400.601-6.009-6.009-6.009-6.009D Probability values for tests of paired differences in means and medians-6.70-6.70-6.400.601-6.009-6.009-6.009-6.009-6.009-6.009-6.009-6.009-6.009-6.009-6.009-6.009-6.009-6.009-6.009-6.009-6.009 </td <th>i</th> <td>CT</td> <td>8.65</td> <td>1.699</td> <td>0.599</td> <td></td> <td>0.016</td>	i	CT	8.65	1.699	0.599		0.016
Disclosers ($N = 680$) GAP $S23$ 0.772 0.772 0.270 $ 0.00DD$ TD TD 1.59 1.717 0.659 -9.09 $0.001BES$ 1.717 0.650 -4.40 $0.001BES$ 1.717 0.620 -4.40 $0.001BES$ is actual EPS reported by IBES. All EPS numbers refer to basic EPS measured on an unadjusted basis b Probability values for tests of paired differences in means and medians b Probability values for tests of paired differences in means and medians c Variable definitions are as follows: P is stock price measured 3 months after the fiscal year-end; $BVPS$ is book value of shareholders' funds per share; EPS^{k} is earnings d Vursue for metric k ($k = GAP$, TD, and IBES EPS); and OCF is operating cash flow per share d vulnes for metric for the fifterence in adjusted P^{2} values from non-nested models. For each disclosure partition (disclosers and non-disclosers), Z-statistics d vulnes for the fifterence in adjusted P^{2} values from non-nested models. For each disclosere partition (disclosers and non-disclosers), Z-statistics d vulnes for P^{2} -dution from P^{2} values from non-nested models. For each disclosere partition (disclosers and non-disclosers), Z-statistics d vulnes for P^{2} -dution for metric PR word with the former for metric PR and PRE model with the former of the metric PR and PRE model with the former of the metric PR and PRE model with the former of PR model of the former of PR model with PRE model with the former of the metric PR model with PRE model PRE model PRE model PRE model with PRE model PRE mod		IBES	30.70	0.584	0.206		0.253
GAP 37.67 0.772 0.772 0.270 $ -$ TDTD 8.23 1.882 0.679 -9.09 0.001 IBES 11.59 11.717 0.620 -9.09 0.001 a Earnings per share (EPS) metrics are as follows: GAAP is the EPS figure mandated by FRS 3; TD is EPS from recurring operations derived by Thomson Datastream; a Earnings per share (EPS) metrics are as follows: GAAP is the EPS figure mandated by FRS 3; TD is EPS from recurring operations derived by Thomson Datastream; a Earnings per share (EPS) metrics are as follows: For basic EPS measured on an unadjusted basis 0.620 -4.40 0.001 b Probability values for tests of paired differences in means and medians b Probability values for tests of paired differences in means and medians c Variable definitions are as follows: P is stock price measured 3 months after the fiscal year-end; $BVPS$ is book value of shareholders' funds per share; EPS^{k} is earnings per share computed for metric k ($k = GAAP$, TD, and IBES EPS); and OCF is operating cash flow per share d Vucuos is 10990 Z-statistic testing the difference in adjusted R^{2} values from non-nested models. For each disclosure partition (disclosers and non-disclosers), Z-statistics cash flow for 0.000 from	5	Disclosers $(N = 680)$					
TD 8.23 1.882 0.679 -9.09 0.001 IBESII.59 1.717 0.620 -9.09 0.001 a Earnings per share (EPS) metrics are as follows: GAAP is the EPS figure mandated by FRS 3; TD is EPS from recurring operations derived by Thomson Datastream; a Earnings per share (EPS) metrics are as follows: GAAP is the EPS figure mandated by FRS 3; TD is EPS from recurring operations derived by Thomson Datastream; b Probability values for tests of paired differences in means and medians b Probability values for tests of paired differences in means and medians c Variable definitions are as follows: P is stock price measured 3 months after the fiscal year-end; BVPS is book value of shareholders' funds per share; EPS ⁴ is earnings per share computed for metric k ($k =$ GAAP, TD, and IBES EPS); and OCF is operating cash flow per share d Vuong's (1989) Z-statistic testing the difference in adjusted R^{2} values from non-nested models. For each disclosere partition (disclosers and non-disclosers), Z-statistics		GAAP	37.67	0.772	0.270		I
IBESII.59I.717 0.620 -4.40 0.001 a Earnings per share (EPS) metrics are as follows: GAAP is the EPS figure mandated by FRS 3; TD is EPS from recurring operations derived by Thomson Datastream;BES is actual EPS reported by IBES. All EPS numbers refer to basic EPS measured on an unadjusted basisb Probability values for tests of paired differences in means and mediansc Variable definitions are as follows: P is stock price measured 3 months after the fiscal year-end; $BVPS$ is book value of shareholders' funds per share; EPS^4 is earnings per share computed for metric k ($k = GAAP$, TD, and IBES EPS); and OCF is operating cash flow per sharea Vuong's (1989) Z-statistic testing the difference in adjusted R^2 values from non-neted models. For each disclosure partition (disclosers and non-disclosers), Z-statistics		CT	8.23	1.882	0.679		0.001
^a Earnings per share (EPS) metrics are as follows: GAAP is the EPS figure mandated by FRS 3; TD is EPS from recurring operations derived by Thomson Datastream; IBES is actual EPS reported by IBES. All EPS numbers refer to basic EPS measured on an unadjusted basis ^b Probability values for tests of paired differences in means and medians ^c Variable definitions are as follows: <i>P</i> is stock price measured 3 months after the fiscal year-end; <i>BVPS</i> is book value of shareholders' funds per share; <i>EPS^k</i> is earnings per share computed for metric <i>k</i> (<i>k</i> = GAAP, TD, and IBES EPS); and OCF is operating cash flow per share ^d Vuong's (1989) Z-statistic testing the difference in adjusted <i>R</i> ² values from non-nested models. For each disclosure partition (disclosers and non-disclosers), Z-statistics		IBES	11.59	1.717	0.620		0.001
IBES is actual EPS reported by IBES. All EPS numbers refer to basic EPS measured on an unadjusted basis ^b Probability values for tests of paired differences in means and medians ^c Variable definitions are as follows: <i>P</i> is stock price measured 3 months after the fiscal year-end; <i>BVPS</i> is book value of shareholders' funds per share; <i>EPS⁴</i> is earnings per share computed for metric k ($k = GAAP$, TD, and IBES EPS); and OCF is operating cash flow per share ^d Vuong's (1989) Z-statistic testing the difference in adjusted R^2 values from non-nested models. For each disclosure partition (disclosers and non-disclosers), Z-statistics common h^2 advinced D^2 advinced from h^0 GAAD EDS model with the correstoring convolute form called or the TD EDS model.	- B	Earnings per share (EPS) metrics are as follow	vs: GAAP is the EPS figure	mandated by FRS 3; TD i	is EPS from recurring operatic	ons derived by Thomson Data	stream;
^c Variable definitions are as follows: <i>P</i> is stock price means and means and means c variable definitions are as follows: <i>P</i> is stock price measured 3 months after the fiscal year-end; <i>BVPS</i> is book value of shareholders' funds per share; <i>EPS</i> ⁴ is earnings per share computed for metric k ($k = GAAP$, TD, and IBES EPS); and OCF is operating cash flow per share ^d Vuong's (1989) Z-statistic testing the difference in adjusted R^{2} values from non-nested models. For each disclosure partition (disclosers and non-disclosers), Z-statistics	I A	BES is actual EPS reported by IBES. All EPS	numbers refer to basic EPS	measured on an unadjust	ed basis		
per share computed for metric k ($k = GAAP$, TD, and IBES EPS); and OCF is operating cash flow per share ^d Vuong's (1989) Z-statistic testing the difference in adjusted R^2 values from non-nested models. For each disclosure partition (disclosers and non-disclosers), Z-statistics common the adjusted P^2 absoluted from the CAAD EDS model with the commendation under from other the IDES EDS model or the TD EDS model	5	Probability values for tests of paired difference Variable definitions are as follows: P is stock p	es in means and medians brice measured 3 months aft	er the fiscal year-end; BVI	PS is book value of shareholde	rrs' funds per share; <i>EPS</i> ^k is e	arnings
^d Vuong's (1989) Z-statistic testing the difference in adjusted R ² values from non-nested models. For each disclosure partition (disclosers and non-disclosers), Z-statistics	<u>ц</u> ,	ber share computed for metric k (k = GAAP, TI	D, and IBES EPS); and OCI	F is operating cash flow p	er share		
		Vuong's (1989) Z-statistic testing the difference waves the adjusted \mathbb{R}^2 obtained from the \mathbb{G}_A	te in adjusted R^2 values from $\Delta D \ FDS \ model \ with the correction$	1 non-nested models. For e	ach disclosure partition (disclored)	osers and non-disclosers), Z-st	tatistics

رات

Thomson recurring earnings as net income minus total non-recurring items identified by Thomson and goodwill amortization.

Panel A of Table 2 reports summary statistics for GAAP, Thomson and IBES EPS. Differences among the three metrics are trivial for non-disclosers (medians differ by less than half a pence). In contrast, GAAP EPS differs statistically and economically from Thomson and IBES EPS for non-GAAP EPS disclosers. The disparity is consistent with the presence of material transitory items in GAAP earnings for firms that disclose non-GAAP EPS. This is confirmed by the evidence reported in panels B and C: GAAP EPS for the discloser group exhibits significantly lower explanatory power for prices (panel B) and one-period-ahead operating cash flow (panel C) compared with our two recurring earnings proxies. Evidence that GAAP earnings lack value and forecasting relevance in the presence of transitory items is consistent with research stretching back to Lipe (1986). By comparison, GAAP earnings for non-disclosers display similar value relevance to the IBES and Thomson metrics, suggesting the absence of large non-recurring items.

If non-GAAP EPS numbers merely adjust for transparently transitory items that financial statement users could easily reverse themselves then the benefits of such disclosures are unclear. Table 3 compares non-GAAP EPS with the Thomson and IBES values. A significant level of agreement over the treatment of transitory items is evident across the three metrics. Management adjustments conform exactly to those made by Thomson (IBES) in 55 (69) percent of cases.⁸ Full consensus between management, Thomson and IBES exists in 41% of all cases where non-GAAP EPS is disclosed. This level of agreement suggests that the task of identifying non-recurring earnings components is often relatively straightforward, presumably because such items are unambiguously transitory. Similar levels of agreement between non-GAAP earnings and IBES earnings have been documented in the United States (Bhattacharya et al. 2003; Marques 2006).

Despite substantial consensus over the identity of non-recurring items, discrepancies between management and external parties are evident in a significant fraction of cases (45% for Thomson and 31% for IBES).⁹ These cases represent uncertainty over the distinction between recurring and transitory earnings components. This disagreement could reflect an attempt by management to mislead investors by misclassifying permanent and transitory items. Alternatively, it could reflect managers' informational advantage in classifying potentially ambiguous earnings components. The remainder of the paper seeks to distinguish between these two competing explanations.

⁹ One could also treat as disagreements those cases where Thomson (IBES) reports adjustments to GAAP earnings but managers do not disclose a non-GAAP figure. Marques (2006) includes such cases in her analysis. We do not follow this route because the focus of our paper is on cases where managers make an explicit decision to report a competing earnings number to GAAP earnings. Although the question of why managers choose *not* to disclose non-GAAP earnings is an interesting one (particularly when transitory items exist), this issue lies beyond the scope of our paper.



⁸ Greater consensus between management and IBES is consistent with (a) IBES analysts' superiority at identifying non-recurring earnings components compared with other sophisticated financial statement users (Brown and Sivakumar 2003) and (b) analysts' reliance on management guidance regarding the incidence and magnitude of non-recurring items.

Three-way EPS comparis	sons ^a	Pairwise c	omparisons ^b		
MAN vs. TD vs. IBES		MAN vs.	TD	MAN vs.	IBES
	Total	Agree	Disagree	Agree	Disagree
MAN = TD = IBES	294	294	_	294	_
MAN = TD \neq IBES	101	101	_	_	101
MAN = IBES \neq TD	199	_	199	199	_
$MAN \neq TD = IBES$	6	_	6	_	6
$MAN \neq TD \neq IBES$	112	_	112	_	112
	712	395	317	493	219

Table 3 Comparison of non-GAAP EPS reported by management with EPS metrics reported by Thomson Datastream and IBES

^a Earnings per share (EPS) metrics are as follows: MAN is the non-GAAP EPS figure reported by management; TD is EPS from recurring operations derived by Thomson Datastream; IBES is actual EPS reported by IBES

^b Comparisons between the value of non-GAAP EPS reported by management and the corresponding values computed by Thomson and IBES, respectively. The columns headed "agree" report the frequencies of cases where MAN = TD and MAN = IBES. The columns headed "disagree" report the frequencies of cases where MAN \neq TD and MAN \neq IBES. All EPS comparisons are performed on an unadjusted basis

4 Disagreement

Discrepancies between non-GAAP EPS and other recurring earnings proxies can occur for two reasons. First, management may exclude *additional* gains and losses beyond those omitted by outsiders. Second, management may choose *not* to exclude certain gains and losses omitted by outsiders. Failure to distinguish between these sources of disagreement confounds management's reporting incentives. To illustrate, consider the case where non-GAAP earnings exceed IBES earnings by \$1 million. On one hand, this discrepancy may arise because management has excluded an additional \$1 million loss beyond IBES. Accordingly, evidence that this adjustment is value relevant suggests classificatory earnings management (that is, a recurring loss has been inappropriately excluded). On the other hand, the \$1 million difference could reflect management reversal of an IBES-excluded gain. In this scenario, evidence that the adjustment is value relevant suggests that management correctly identified and included a recurring gain that IBES misclassified as transitory. We therefore disaggregate disagreements into management inclusions and exclusions.

4.1 Disaggregation method

Identifying the sources of disagreement between management and outsiders requires complete information on adjustments made by both parties. FRS 3's requirement for a full reconciliation between GAAP and non-GAAP EPS ensures transparency of management's adjustments. IBES EPS represents an obvious candidate against which to compare management's adjustments. Unfortunately, IBES does not



provide information on analysts' individual adjustments to GAAP earnings, making it impossible to reconcile cases of disagreement on an item-by-item basis. As an alternative we focus on Thomson's measure of recurring earnings.

For the 317 cases of disagreement between the non-GAAP and Thomson earnings metrics reported in Table 3, we use the following procedure to reconcile earnings components classified by Thomson as non-recurring with items excluded from GAAP EPS by management. We first map aggregate non-recurring operating and non-operating items reported by Thomson onto individual income statement items using footnote disclosures. The resulting list of items classified by Thomson as non-recurring is then compared with the list of items excluded from GAAP EPS by management using reconciliation disclosures mandated by FRS 3. Items in both lists represent transitory items over which management and Thomson agree, while transactions unique to one list represent cases of disagreement. Items unique to Thomson's list are transactions excluded by Thomson but included by management in its non-GAAP figure. We label these items "management inclusions" and distinguish between included gains and losses. Conversely, items unique to management's list are transactions excluded by management but not Thomson. We label these items "management exclusions," again distinguishing between gains and losses.¹⁰ Further details of our reconciliation method are provided in an appendix. Our approach is similar to Gu and Chen (2004), who distinguish between transitory earnings components included and excluded by First Call analysts.

4.2 Summary statistics

Panel A of Table 4 presents descriptive statistics on the items about which management and Thomson disagree. The median value for net adjustments (management exclusions minus management inclusions) is positive: non-GAAP EPS is typically less than Thomson EPS in the presence of disagreement. However, net adjustments are left-skewed by cases where non-GAAP EPS exceeds Thomson EPS by a relatively large amount. (Further analysis of panel A reveals that this left-skewness is due to instances where management excludes very large losses beyond Thomson.) Decomposing net adjustments reveals that managers tend to reverse Thomson exclusions almost twice as often as they exclude additional items beyond Thomson. Loss items represent the primary source of disagreement. Of the 236 cases where managers reverse Thomson exclusions, 168 (71%) relate to losses.

¹⁰ Two further practical considerations concerning the reconciliation process are worthy of note. The first involves the treatment of taxation. While Thomson reports all non-recurring items on a pre-tax basis, the approach used by firms is mixed: some report reconciliation items on a pre-tax basis with the aggregate tax effect disclosed separately while others report reconciliation items net of tax. The task of reconciling Thomson adjustments with management adjustments is relatively straightforward when the tax treatment is consistent. For cases where the treatment differs, we restated management exclusions to a pre-tax basis using footnote disclosures. The second complication involves the treatment of goodwill amortization. While non-GAAP EPS often excludes goodwill amortization, Thomson's recurring earnings metric is computed *before* deducting goodwill. To ensure comparability between the two metrics, we adjusted the Thomson number to a pre-goodwill amortization basis using the adjustments reported by management. This procedure ensures that in no case does the treatment of goodwill amortization account for disagreement between Thomson and management.



(amine ind amine) containing aniderican interior		(
	Incremental managemen	management adjustments beyond Thomson	nd Thomson				
	Net adjustments ^a	Inclusions			Exclusions		
		Gains	Losses	Total	Gains	Losses	Total
Mean	-0.018	1.321	-2.705	-1.570	2.683	-5.521	-3.007
Std	7.636	1.915	7.889	6.994	4.591	9.278	8.958
Median	0.091	0.758	-0.887	-0.260	1.158	-1.721	-0.814
N^{p}	317	66	168	236	38	86	124
					% of firms ^c		Z-stat for difference ^d
Earnings component	ponent				Agree	Disagree	
Management	Management exclusions $(N = 124)$						
Non-operating activities	g activities						
Profit (los	Profit (loss) on sale or termination of operations	rations			56.6	1.6	-11.15
Profit (los	Profit (loss) on sale of fixed assets or investments	estments			34.2	11.3	-5.05
Reorganiz	Reorganization, restructuring and integration costs	ion costs			37.4	4.8	-7.09
Other spec	Other special gains (losses)				15.3	0.8	-4.40
Other non-	Other non-operating adjustments and adjustments to associate profits	istments to associate	profits		8.0	2.4	-2.22
Exception	Exceptional dividend, interest or taxation				0.7	14.5	8.05
Operating activities	ivities						
Impairmen	Impairment, diminution or write-off of goodwill	lliwpoo			2.6	6.5	2.22
Revaluatio	Develoption of immediation of final accets of investments				č	L C T	

608

ت

ارات				
	Table 4 continued			
		$\%$ of firms $^{\rm c}$		Z-stat for difference ^d
W	Earnings component	Agree	Disagree	
2	Acquisition, merger and demerger costs	3.7	4.0	0.16
i	Other specified exceptional costs	15.5	20.2	1.29
J	Other unspecified exceptional items	4.3	9.7	2.45
	Other			
	Discontinued operations	0.8	21.0	10.06
ł	Provisions	2.9	8.1	2.75
	Adjustment of weighted no. of shares	0.0	4.8	5.33
5	Management inclusions $(N = 236)$			
	Non-operating activities			
	Profit (loss) on sale or termination of operations	68.6	6.8	-15.50
	Profit (loss) on sale of fixed assets or investments	40.6	12.8	-7.50
	Reorganization, restructuring and integration costs	9.2	1.7	-3.75
	Other special gains (losses)	19.9	5.6	-4.99
	Other non-operating adjustments and adjustments to associate profits	11.9	20.5	3.04
	Operating activities ^c	57.3	71.4	3.64
	^a Net adjustments are defined as exclusions minus inclusions. A positive (negative) value for net adjustments means that non-GAAP earnings are lower (higher) than Thomson recurring earnings	nts means that no	on-GAAP earning	gs are lower (higher) than
	^b The sum of total inclusions (236) plus total exclusions (124) exceeds the total number of disagreement cases because some firms make both types of incremental adjustment	ecause some firm	s make both types	of incremental adjustment
	^c For exclusions, 124 cases (81 exclusion-only cases plus 43 cases involving both exclusions and inclusions) involve management excluding additional items beyond Thomson (Disarree subsample) compared with 588 where management exclusions are replicated by Thomson (Arree subsample). For inclusions, 236 cases (193	ns) involve mana omson (Agree su	gement excluding ibsample). For it	g additional items beyond nclusions, 236 cases (193
<u></u>	inclusion-only cases plus 43 cases involving both inclusions and exclusions) involve management reversals of Thomson exclusions (Disagree subsample) compared with 476 where management do not add back Thomson exclusions (Agree subsample). Column totals exceed 100% because managers adjust for multiple items	f Thomson exclu 0% because man	isions (Disagree s agers adjust for 1	ubsample) compared with multiple items
prinț	^d Z-stat for the difference in sample proportions between Agree and Disagree relate to a proportional Z- test (Newbold, 1988)	st (Newbold, 198	(8)	1

Disagreement over the persistence of earnings components

609

^e Disaggregation of operating inclusions into individual components is not possible in all cases due to insufficient disclosure

Further, included losses tend to be larger in magnitude relative to included gains. These results do not support the view that disagreement reflects management efforts to present firm performance in a more favorable light.

Conversely, 86 out of 124 management-specific exclusions (70%) involve losses, and the magnitude of the mean (median) excluded loss exceeds the corresponding value for excluded gains. These adjustments are consistent with management seeking to boost reported earnings by classifying recurring losses as transitory (McVay 2006). Alternatively, these adjustments may represent appropriate exclusion of transitory losses that Thomson misclassifies as recurring.

Panel B of Table 4 presents evidence on earnings-component adjustments about which management and Thomson either agree or disagree. Of the 712 cases where non-GAAP earnings are disclosed, 124 involve management excluding additional items beyond Thomson (Disagree), compared with 588 where management exclusions are replicated by Thomson (Agree). The top part of panel B reports the percentage of observations in each subsample where a particular earnings component is excluded. Differences in exclusion percentages are evidence that earnings component exclusions are not evenly distributed across the two subsamples. (Column totals sum to greater than 100% because managers often exclude more than one item.) For example, gains and losses on termination of operations are excluded from GAAP earnings by management and Thomson in almost 57% of agreement cases. In contrast, the same item accounts for less than 2% of cases where managers make incremental exclusions beyond Thomson [difference in proportions is significant at the 1% level using a proportional Z-test (Newbold 1988)]. Disagreement over the treatment of gains and losses on termination of operations therefore appears to be a relatively rare event in the U.K. With the exception of the dividend, interest and taxation item, all other non-operating categories are also significantly underrepresented in the disagreement subsample. One explanation for this pattern is that the underlying nature of most non-operating items, coupled with FRS 3's transparent disclosure requirements, creates little room for confusion over the persistence of such components. Instead, a disproportionate number of exclusion disagreements appear to center on operating earnings components whose persistence is difficult to determine due to their idiosyncratic nature or poor disclosure (for example, other unspecified exceptional operating items). Discontinued operations also account for a disproportionately large number of disagreement cases, most likely reflecting ambiguities caused by FRS 3's restricted definition of what discontinued activities may comprise.

The bottom part of panel B presents a similar analysis for management inclusions. Of the 712 cases where non-GAAP earnings are disclosed, 236 involve management reversals of Thomson exclusions (Disagree), compared with 476 where management do not add back Thomson exclusions (Agree). Only rarely do managers reverse Thomson-excluded non-operating items. Instead, operating items account for the majority of reversals. Overall, panel B indicates that disputes between management and Thomson are more likely to reflect idiosyncratic earnings components whose persistence is difficult for outsiders to gauge, opaque disclosure (non-recurring operating items), or restrictive classification rules (discontinued operations).



5 Properties of contested items

This section presents results of tests exploring management's reporting incentives in the event of disagreement. We conduct valuation and predictability tests designed to shed light on the informativeness of incremental management adjustments beyond Thomson. Our tests are based on the following decomposition of GAAP earnings:

$$GAAPEPS \equiv MANEPS + TNONREC + MANEXCL - MANINCL + GTM$$
(1)

where *GAAPEPS* is EPS under FRS 3, *MANEPS* is non-GAAP earnings disclosed by management, *TNONREC* is total non-recurring items identified by Thomson, *MANEXCL* is additional aggregate management exclusions beyond *TNONREC*, and *MANINCL* is aggregate items classified by Thomson as non-recurring but included by management in *MANEPS*. All non-recurring items are measured on a pre-tax basis. Finally, *GTM* represents the net of goodwill amortization excluded by management, plus aggregate tax and minority interests on excluded earnings components.

5.1 Incremental value and forecasting relevance

We examine the incremental value relevance of disputed items by substituting Eq. 1 into a standard book value and earnings valuation regression (Collins et al. 1997):

$$Price_{it} = \delta_0 + \delta_1 BVPS + \delta_2 MANEPS_{it} + \delta_3 TNONREC_{it} + \delta_4 MANEXCL_{it} - \delta_5 MANINCL_{it} + \delta_6 GTM_{it} + v_{it}$$
(2)

where *Price* is stock price per share measured 3 months after the fiscal year-end; *BVPS* is book value of shareholders' funds per share; *i* and *t* are firm and time subscripts, respectively; *v* is the regression residual; and all remaining variables are as defined in Eq. 1 and measured on a per share basis. All variables are scaled by lagged price to reduce heteroskedasticity.¹¹ If incremental adjustments are driven by management's superior knowledge of recurring and non-recurring items then the estimated coefficient on *MANINCL* in model (2) will be negative and significant while the coefficient on *MANEXCL* will be value irrelevant. Conversely, if strategic disclosure drives these adjustments then *MANEXCL* will be value relevant whereas *MANINCL* will be insignificant. (Assuming that opportunism is best reflected in income-increasing adjustments, such effects should be especially apparent for excluded losses and included gains.)

We also examine the incremental forecasting relevance of management exclusions and inclusions using the following prediction model for 1-year-ahead and aggregate 2-year-ahead operating cash flow:

$$CFO_{it+n} = \gamma_0 + \gamma_1 MANEPS_{it} + \gamma_2 TNONREC_{it} + \gamma_3 MANEXCL_{it} - \gamma_4 MANINCL_{it} + \gamma_5 GTM_{it} + \mu_{it}$$
(3)

¹¹ We also estimated all models using unscaled data (Barth and Kallapur 1996). None of the key inferences are affected.



where *CFO* is operating cash flow per share measured over period t + n (where *n* is either one year or aggregate two years); μ is the regression residual; and all remaining variables are as defined in regression (2). If management correctly excludes additional non-recurring items overlooked by Thomson then *MANEXCL* should have no ability to forecast future cash flow. If management correctly reverses Thomson exclusions of recurring items then *MANINCL* should display predictive ability.

Results for models (2) and (3) are reported in Table 5. The top and bottom 1% of observations are excluded in each model to reduce the impact of outliers. Valuation tests are reported in columns 2-5. Coefficient estimates on BVPS and MANEPS are positive and significant in all models as expected. The estimated coefficient on TNONREC is statistically zero in M1, consistent with Thomson correctly classifying these items as non-recurring. Incremental management exclusions are also insignificant, suggesting that on average managers identify additional transitory earnings components overlooked by Thomson analysts. Conversely, management inclusions are incrementally value relevant (two-tailed *p*-value < 0.05). Moreover, the coefficient estimate on MANINCL (-4.26) is similar in magnitude to the multiple on MANEPS (4.99).¹² These findings suggest that managers correctly identify and retain recurring earnings components misclassified by Thomson as transitory. Model M2 disaggregates total Thomson non-recurring items into operating and non-operating elements. Non-operating items are insignificant. However, the coefficient on operating activities is positive and significant, suggesting that Thomson incorrectly classifies some recurring operations as transitory. The significant coefficient estimate on MANINCL reflects this mistake: managers correctly retain recurring items in non-GAAP EPS that Thomson mistakenly considers transitory.

Models M3 and M4 decompose management-specific inclusions and exclusions into net gains and losses. Although excluded gains are marginally significant, the positive coefficient estimates support neither the opportunistic nor informative view. Excluded losses are insignificant and the point estimates are close to one, consistent with these items being transitory. Included losses are incrementally value relevant in both models with coefficient estimates similar in magnitude to *MANEPS*. Management correctly includes recurring losses that Thomson mistakenly considers non-recurring. These findings support the view that non-GAAP EPS disclosures reflect management's superiority at partitioning earnings components into recurring and non-recurring categories. Nevertheless, evidence consistent with strategic disclosure of non-GAAP EPS is apparent for included gains, where the valuation multiple is statistically insignificant. These results suggest that management sometimes retains transitory gains in *MANEPS* in an effort to boost reported performance.

The remaining columns in Table 5 report predictability results. Findings are generally consistent with the valuation results. In aggregate, management-specific exclusions have no incremental explanatory power for future operating cash flows

 $^{^{12}}$ From Eq. 2 and consistent with Marques (2006), the expected coefficient sign on *MANINCL* is negative for included items that are value relevant.



1.41	Table 5 Coefficient estimates and model summary statistics for incremental value relevance tests of management-specific adjustments to non-GAAP EPS	CICILI CONTINUA)				
	Variables ^c	Valuation models ^a	models ^a			Predictabi	Predictability models (1yr ahead) ^b	(1yr ahead) ^b		Predictabi	Predictability models (aggr. 2yr ahead) ^b	ggr. 2yr ahead	q(
SIL		MI	M2	M3	M4	M5	M6	M7	M8	9M	M10	M11	M12
*	Intercept	0.705 (0.001)	0.714 (0.001)	0.693 (0.001)	0.696 (0.001)	0.034 (0.001)	0.036 (0.001)	0.032 (0.001)	0.034 (0.001)	0.085 (0.003)	0.088 (0.004)	0.079 (0.009)	0.085 (0.006)
	BVPS	0.116 (0.015)	0.114 (0.023)	0.134 (0.005)	0.128 (0.010)								
V	MANEPS	4.989 (0.001)	5.131 (0.001)	4.870 (0.001)	4.999 (0.001)	1.673 (0.001)	1.678 (0.001)	1.685 (0.001)	1.669 (0.001)	3.367 (0.001)	3.373 (0.001)	3.428 (0.001)	3.382 (0.001)
1	TNONREC	0.320 (0.319)		0.429 (0.159)		-0.020 (0.773)		0.012 (0.858)		-0.124 (0.501)		0.013 (0.942)	
	TDEXC_OP		1.928 (0.004)		1.415 (0.014)		0.232 (0.102)		0.098 (0.439)		0.196 (0.688)		0.290 (0.549)
1	rDEXC_NOP		0.220 (0.627)		0.346 (0.435)		0.017 (0.868)		0.079 (0.444)		0.157 (0.576)		0.351 (0.202)
V	MANEXCL	2.501 (0.150)	2.191 (0.213)			0.262 (0.515)	0.133 (0.743)			0.767 (0.492)	0.330 (0.769)		
V	MANEXCL ⁺			8.383 (0.051)	8.975 (0.039)			-0.154 (0.869)	-0.104 (0.912)			2.974 (0.390)	3.819 (0.276)
V	MANEXCL ⁻			1.345 (0.225)	1.289 (0.251)			-0.043 (0.865)	-0.097 (0.699)			0.145 (0.815)	-0.058 (0.926)
V	MANINCL	-4.257 (0.015)	-5.107 (0.006)			-1.856 (0.001)	-2.031 (0.001)			-3.499 (0.007)	-3.563 (0.008)		
V	MANINCL ⁺			4.820 (0.171)	4.514 (0.380)			0.989 (0.202)	1.101 (0.344)			2.414 (0.252)	2.220 (0.527)

D Springer

<u>@</u>	Table 5	Table 5 continued											
Sprin	Variables ^c		Valuation models ^a			Predictal	bility models	Predictability models (1yr ahead) ^b	٩	Predictab	ility models (Predictability models (aggr. 2yr ahead) ^b	1) ^b
ger		MI	M2	M3	M4	M5	M6	M7	M8	6M	M10	M11	M12
Ä	MANINCL	CL^{-}		-4.058 (0.012)	-4.845 (0.004)			-1.663 (0.001)	-1.720 (0.001)			-2.978 (0.006)	-3.060 (0.007)
JL	GTM	4.361 (0.001)	1 3.402 1) (0.015)	4.152 (0.001)	3.017 (0.027)	0.443 (0.121)	0.465 (0.123)	0.277 (0.323)	0.331 (0.275)	0.968 (0.306)	1.260 (0.244)	0.880 (0.353)	1.625 (0.135)
	Adj R^2	0.343	3 0.349	0.355	0.360	0.457	0.463	0.437	0.438	0.295	0.296	0.295	0.300
	N	291	288	299	295	276	273	284	280	212	209	218	214
ľ	Tests ard probabil	Tests are based on the 317 cases probability values are reported in		ere the non-G/ rentheses	AAP EPS figui	re reported	by managen	nent differs fi	rom recurring	g EPS comp	uted by Thom	where the non-GAAP EPS figure reported by management differs from recurring EPS computed by Thomson Datastream. Two-tailed 1 parentheses	ı. Two-tailed
	^a The d. observat	^a The dependent variable in the observations for each variable an		ation models cluded to redu	is price per sluce the impact	nare, measu of outliers	ired 3 montl	hs after the fi	iscal year-end	l, scaled by	lagged price.	valuation models is price per share, measured 3 months after the fiscal year-end, scaled by lagged price. The top and bottom 1% of e excluded to reduce the impact of outliers	ottom 1% of
	^b The du and bott	^b The dependent variable is 1-ye and bottom 1% of observations	ble is 1-year-al servations for e	ara-ahead (1yr ahead) or aggregate one plus 2-years-ahead (aggr. for each variable are excluded to reduce the impact of outliers	 d) or aggregate re excluded to 	e one plus 2 reduce the	P-years-ahead	d (aggr. 2yr al outliers	head) operatir	ng cash flow	per share, sca	The dependent variable is 1-year-ahead (1yr ahead) or aggregate one plus 2-years-ahead (aggr. 2yr ahead) operating cash flow per share, scaled by lagged price. The top and bottom 1% of observations for each variable are excluded to reduce the impact of outliers	rice. The top
	 varial non-rect aggregat excludec are meas 	^c Variable definitions are as foll non-recurring items classified b aggregate items included by ma excluded by management from C are measured on a per share bas	^c Variable definitions are as follows: BVPS is book value of non-recurring items classified by Thomson; <i>MANEXCL</i> is agaggregate items included by management in <i>MANEPS</i> despit excluded by management from GAAP EPS, aggregate tax on a are measured on a per share basis and scaled by lagged price	: BVPS is boo iomson; MANI ment in MAN. P EPS, aggreg d scaled by la	k value of sha <i>EXCL</i> is aggre <i>EPS</i> despite th ate tax on all ϵ gged price	reholders' sgate addit nese items sarnings co	funds; <i>MAN</i> ional items being classi mponents ex	<i>EPS</i> is non-C excluded by fied in <i>TNOA</i> ccluded by ccluded by ccluded by eit	AAP earning management <i>REC</i> by Thc her managen	gs disclosed from GAAl mson; and v nent or Thon	by managem P EPS beyon <i>GTM</i> is repre- nson, and min	^c Variable definitions are as follows: BVPS is book value of shareholders' funds; <i>MANEPS</i> is non-GAAP earnings disclosed by management; <i>TNONREC</i> is aggregate non-recurring items classified by Thomson; <i>MANEXCL</i> is aggregate additional items excluded by management from GAAP EPS beyond <i>TNONREC</i> ; <i>MANINCL</i> is aggregate items included by management from GAAP EPS beyond <i>TNONREC</i> ; <i>MANINCL</i> is aggregate items included by management from GAAP EPS, and <i>TNONREC</i> ; <i>MANINCL</i> is excluded by management from GAAP EPS, aggregate these items being classified in <i>TNONREC</i> by Thomson; and <i>GTM</i> is represents goodwill amortization excluded by management from GAAP EPS, aggregate tax on all earnings components excluded by either management or Thomson, and minority interests. All variables are measured on a per share basis and scaled by lagged price	is aggregate MANINCL is amortization All variables

Springer ۩ارات

615

whereas the coefficient on *MANINCL* is significant and similar in magnitude to the multiple on *MANEPS*. Decomposing exclusions, neither gains nor losses are significant, suggesting that both components are transitory. Included losses on the other hand are highly significant suggesting that these items represent recurring earnings components. Once again, these results suggest that management-specific adjustments reflect insiders' superior knowledge concerning the true persistence of earnings components. As with our valuation results, however, opportunistic considerations are more consistent with inclusion of forecast irrelevant incremental gains.

Overall, findings reported in Table 5 suggest that management-specific adjustments to GAAP earnings more likely reflect insiders' superior knowledge of the persistence of earnings components.¹³ However, when aggregate management inclusions are further decomposed into gains and losses, evidence of opportunism emerges in relation to the treatment of transitory gains. Therefore, while U.K. non-GAAP earnings disclosures typically reflect management's ability to identify recurring components of net income, a significant minority of adjustments appear to be driven by efforts to present firm performance in a more favorable light.

5.2 Conditioning on IBES adjustments

Net disagreements between management and Thomson may be decomposed into adjustments that correspond with IBES and those that conflict with IBES. Marques (2006) finds that investors typically view IBES adjustments as an appropriate elimination of transitory items. Given the high level of correspondence between non-GAAP earnings and IBES actual earnings in our sample (69%, Table 3), it is possible that the results in Table 5 are driven by the subset of incremental management adjustments that match those made by IBES. In other words, management-specific adjustments may be adding little information beyond IBES.

We address this issue by constructing an indicator variable (*IBESADJ*) that takes the value of one where non-GAAP EPS equals IBES actual EPS and zero otherwise. *IBESADJ* captures the subset of management-Thomson disputes where management's adjustments match those made by IBES. We then interact*IBESADJ* with *MANEXCL* and *MANINCL* in regressions (2) and (3). The main effect coefficients on *MANEXCL* and *MANINCL* in these extended models capture the value and forecasting relevance of management-specific adjustments that differ from IBES adjustments, while the *MANEXCL* × *IBESADJ* and *MANINCL* × *IBESADJ* interaction terms capture the incremental effects associated with IBES-consistent adjustments.

¹³ We examined the robustness of these results across sample years. Since the number of observations available for separate annual regressions is small, we reestimated the models reported in Table 5 using combinations of any two sample-years (i.e., 1993/4 and 1996, 1993/4 and 2001, and 1996 and 2001). Untabulated regression results for all 2-year combinations are consistent with those reported in Table 5 using the full sample. In particular, the estimated coefficient on *MANEXCL* is always insignificant, whereas *MANINCL* is always negative and generally significant. Since 1993/4 was the first year under FRS 3, we also tested for the presence of adoption-period effects by interacting the management inclusion and exclusion variables with a 1993/4 indicator variable. No systematic differences for the 1993/4 period are apparent.



Variable ^c	Valuation model ^a	Predictability model	s ^b
		1-yr CFO	Agg. 2-yr CFO
Intercept	0.699 (0.001)	0.033 (0.001)	0.085 (0.004)
BVPS	0.116 (0.015)		
MANEPS	5.079 (0.001)	1.698 (0.001)	3.395 (0.001)
TNONREC	0.309 (0.337)	-0.029 (0.680)	-0.111 (0.553)
MANEXCL	3.210 (0.142)	0.197 (0.681)	0.573 (0.687)
MANEXCL × IBESADJ	-0.900 (0.797)	0.448 (0.591)	0.646 (0.777)
MANINCL	-8.138 (0.007)	-2.651 (0.001)	-4.358 (0.010)
MANINCL × IBESADJ	5.643 (0.112)	1.170 (0.125)	2.011 (0.410)
GTM	4.208 (0.002)	0.456 (0.117)	1.025 (0.300)
Adj R^2	0.344	0.459	0.291
Ν	291	276	212

 Table 6
 Evidence on disagreements between management and Thomson conditional on IBES adjustments

Tests are based on the 317 cases where the non-GAAP EPS figure reported by management differs from recurring EPS computed by Thomson Datastream. Two-tailed probability values are reported in parentheses

^a The dependent variable in the valuation model is price per share, measured 3 months after the fiscal year-end, scaled by lagged price. The top and bottom 1% of observations for each variable are excluded to reduce the impact of outliers

^b The dependent variable is 1-year-ahead (1yr CFO) or aggregate one plus 2-years-ahead (Agg. 2yr CFO) operating cash flow per share, scaled by lagged price. The top and bottom 1% of observations for each variable are excluded to reduce the impact of outliers

^c *IBESADJ* is an indicator variable taking the value of one where *MANEPS* is equal to IBES actual EPS and zero otherwise. All remaining variables are defined in Table 5. All variables are measured on a per share basis and scaled by lagged price

Results reported in Table 6 suggest that management adjustments beyond IBES are informative about the persistence of earnings components. The *MANEXCL* and *MANINCL* main effect coefficient estimates capturing adjustments by management that differ from IBES are entirely consistent with those in Table 5, while the interaction effects capturing the incremental properties of IBES-consistent adjustments are insignificant at the 5% level. Management adjustments beyond Thomson are therefore associated with similar value and forecasting relevance irrespective of whether they correspond to adjustments made by IBES.

5.3 Relative value and forecasting relevance

If management adjustments beyond Thomson are informative about recurring earnings on average, then non-GAAP earnings should be more representative of permanent earnings than the corresponding Thomson number. We compare the relative value relevance of non-GAAP earnings and Thomson recurring earnings to determine which measure has the higher explanatory power for prices. Following Brown and Sivakumar (2003), we modify the standard book value and earnings valuation model by decomposing net income into recurring earnings and residual components:



$$Price_{it} = \varpi_0 + \varpi_1 BVPS + \varpi_2 RECUR_{it} + \varpi_3 (NI_{it} - RECUR_{it}) + \xi_{it}, \qquad (4)$$

where *Price* is stock price per share measured 3 months after the fiscal year-end; *BVPS* is book value of shareholders' funds per share; *RECUR* is either non-GAAP EPS or Thomson recurring EPS; *NI* is GAAP net income per share according to FRS 3; *i* and *t* are firm and time subscripts, respectively; ξ is the regression residual. We compare the adjusted R^2 values from regression (4) estimated for each of our two recurring earnings proxies using a Vuong (1989) test. We also compare the coefficient estimates on *RECUR* using a *t*-test to determine if either earnings measure is associated with a higher valuation multiplier (Collins et al. 1997; Brown and Sivakumar 2003). We also use a similar approach to assess the relative ability of each recurring earnings measure to explain one-period-ahead operating cash flow.

Results reported in Table 7 provide no evidence that management-specific adjustments systematically obscure the distinction between transitory and recurring earnings components: in no case is the adjusted R^2 or multiplier for non-GAAP earnings *less* than the corresponding Thomson value. Instead, non-GAAP earnings explain greater variation in prices (p < 0.05) and future operating cash flow (p < 0.10) compared with Thomson recurring earnings. The valuation and forecasting multipliers on non-GAAP earnings are also larger than the corresponding multipliers on Thomson recurring earnings (p < 0.10). These results, although statistically moderate, provide further evidence that disputes between management and Thomson over adjustments to GAAP income typically reflect management's

1 0						
	Ν	Intercept	BVPS	RECUR	NI-RECUR	Adj-R ²
Valuation model ^a						
MAN	292	0.690	0.144	5.246	0.866	0.293
TD	292	0.768	0.156	4.101	0.755	0.255
p-value for difference ^b				0.081		0.036
Forecasting model ^a						
MAN	288	0.034		1.835	0.047	0.317
TD	288	0.064		1.427	-0.028	0.258
p-value for difference ^b				0.028		0.062

 Table 7
 Evidence on the relative value and forecasting relevance of non-GAAP and Thomson earnings in the presence of disagreement

^a The dependent variable in the valuation model is price per share, measured 3 months after the fiscal year-end, scaled by lagged price. The dependent variable in the forecasting model is operating cash flow per share scaled by lagged price. Explanatory variables, all scaled by lagged price, are as follows: *BVPS* is book value of shareholders' equity per share; *RECUR* is either non-GAAP EPS (MAN) or Thomson recurring EPS (TD); and *NI-RECUR* is the difference between net income per share and either non-GAAP EPS or Thomson recurring EPS. The top and bottom 1% of observations for all variables in each regression model are excluded to reduce the impact of outliers

^b Probability values for *RECUR* relate to a *t*-test comparing the difference in regression coefficients between the model estimated using non-GAAP EPS and the corresponding model estimated using Thomson recurring EPS. Probability values for $Adj-R^2$ refer to a Vuong test comparing the difference in adjusted R^2 values between the model estimated with non-GAAP EPS and the corresponding model estimated with Thomson recurring EPS.

superior ability to isolate recurring earnings components. The findings also suggest that any reduction in value and forecasting relevance due to strategic inclusion of transitory gains (see previous section) is outweighed by the inclusion of recurring losses and the exclusion of additional transitory items.

6 Summary and conclusions

Prior research demonstrates that non-GAAP earnings reported by management often conform to sustainable earnings proxies derived by analysts. If non-GAAP earnings merely adjust for earnings components that are unambiguously transitory then the costs and benefits of such disclosures are unclear. However disparities between adjustments made by management and analysts do occur in a substantial number of cases. These cases are interesting because they reflect disagreement over the persistence of earnings components. On the one hand, disputed items could reflect attempts by management to mislead investors by misclassifying permanent and transitory items. Alternatively, the disagreement could reflect managers' informational advantage in classifying ambiguous earnings components.

We examine the causes of disagreement about the persistence of earnings components by comparing non-GAAP earnings in the U.K. with Thomson Datastream's measure of recurring earnings. Where the two earnings metrics differ, we use mandated income statement and footnote disclosures to identify the source and properties of disputed items. We find that disagreements are not randomly distributed across earnings components. Whereas non-operating items represent a large fraction of all adjustments made to GAAP earnings by management and Thomson, they account for a disproportionately small fraction of disputed cases. Instead, disagreements occur more frequently for operating items and discontinued operations where the idiosyncratic nature of gains and losses, combined with opaque disclosure, create room for confusion over the persistence of certain items.

Incremental value and forecasting relevance tests suggest that the majority of management-specific adjustments reflect appropriate classification of earnings components by insiders. Nevertheless, evidence consistent with strategic disclosure does emerge for the subset of adjustments where managers reverse Thomson-excluded gains. However, such cases represent less than 10% of the non-GAAP disclosure sample and only 20% of disputed adjustment cases, indicating that although some firms appear to use non-GAAP earnings disclosures to artificially boost reported performance, opportunism does not appear to characterize our data on average. This conclusion is supported by relative value and forecasting relevance tests, which indicate that any reduction in value and forecasting relevance due to strategic inclusion of transitory gains is outweighed by the inclusion of recurring losses and the exclusion of additional transitory items.

Our findings make several contributions to the literature on non-GAAP earnings reporting. First, we focus on the subset of disputed adjustments. These cases are interesting because they enable us to construct relatively sharp tests of management's reporting incentives. Second, we extend Marques (2006) by identifying the specific sources of disagreement and examining the properties of each disputed



component separately. Like Marques (2006), we find some evidence of opportunistic reporting. But more often management-specific adjustments reflect appropriate classification of earnings components. Third, we extend work by Walker and Louvari (2003) on non-GAAP reporting in the United Kingdom by examining the properties of management adjustments relative to adjustments made by other informed parties.

Acknowledgements We thank IBES International for providing IBES actual EPS data. We gratefully acknowledge helpful comments and suggestions from David Citron, Andrew Jones, Paul Klumpes, Stephen Penman (the editor) and two anonymous reviewers, as well as from seminar participants at Lancaster, the London School of Economics, Warwick, the 2004 Financial Reporting and Business Communication Conference (Cardiff), and the 2005 Annual Meetings of the American Accounting Association (San Francisco). We also thank Valentine Ururuka for research assistance. Generous financial support has been provided by the Institute of Chartered Accountants in England and Wales' Centre for Business Performance. A previous version of this paper was titled "Voluntary disclosure of adjusted earnings per share numbers".

Appendix: description of reconciliation method

Non-GAAP earnings numbers are hand-collected from firms' financial statements and reconciled with Thomson Datastream recurring earnings (item #210). Item #210 equals GAAP net income (#625) minus total non-recurring items (net of tax and minority interests) identified by Thomson (#194). Thomson decomposes #194 as follows:

$$#194 = [(#1079 - #1080 + #1081 + #1082 + #1091 - #1090) - #981] + (#989 - #1097)$$

where #1079 is gains and losses on termination of operations; #1080 is reorganization and restructuring costs; #1081 is gains and losses on the sale of fixed assets; #1082 is other special gains and losses; #1091 is other non-operating adjustments; #1090 is adjustments to associate profits; #981 is net non-recurring operating losses before tax; #989 is total tax adjustments; and #1097 is minority interests in non-operating items.

Items in [.] in Eq. A1 are compared with management's adjustments to GAAP earnings using mandated footnote disclosures reconciling non-GAAP earnings with earnings computed under GAAP. Disagreement between management and Thomson over the classification of earnings components occurs when the values reported in management's list of adjustments to GAAP earnings do not match those contained in Thomson's list.

Incremental inclusions by management

Items unique to Thomson's list of non-recurring items represent transactions that management elected *not* to exclude from GAAP earnings despite Thomson having

(A1)

classified the item as transitory (that is, management effectively reverse Thomson adjustments). We label these as "management inclusions," an example of which is illustrated by Nurdin and Peacock PLC for the financial year ending December 31 (all figures in £000):

Non-GAAP earnings			Thomson recurring e	arnings	
GAAP earnings		21877	Item #625		21877
Less Thomson operating items (pre-tax)					
Net deficit on revaluation	-147				
Permanent diminution in fixed asset value	-779				
Provision for compensation for loss of office	-346				
Total non-recurring operating items		(-1272)	Item #981 (× - 1)		(-1272)
Less Thomson non-operating items (pre-tax)					
			Item #1079	0	
			Item #1080 ($\times - 1$)	0	
Loss on sale of fixed assets	-550		Item #1081	-550	
			Item #1082	0	
Gain on sale of current asset investment	438		Item #1091 – #1090	438	
		(-112)			(-112)
Less: tax and minority interests		(660)	Item #989 - #1097		(660)
Thomson non-recurring earnings		22601	Item #210		22601
Management reversal of Thomson exclusions					
Provision for compensation for loss of office	-346				
Gain on sale of current asset investment	438				
Aggregate management inclusions		92			
Reversal of tax and minority interests effects		479			
Non-GAAP earnings		23172			

Incremental exclusions by management

Items unique to management's list represent components excluded from GAAP earnings by management *in addition* to items excluded by Thomson. We label these "management exclusions", an example of which is illustrated by W. H. Smith Group PLC for the financial year ending June 1, 1996 (all figures in £000):



Non-GAAP earnings			Thomson recurring earnings		
GAAP earnings		-200300	Item #625		-200300
Less agreed operating items (pre-tax	x)				
Redundancy and associated costs	-16600				
Fixed asset write-offs	-9400				
Property costs	-30400				
Other operating costs	-24600				
Total non-recurring operating items		(-81000)	Item #981 (× - 1)		(-81000)
Less agreed non-operating items (pr	e-tax)				
Loss on sale of operations	-151300		Item #1079	-151300	
			Item #1080	0	
			(x - 1)		
Loss on sale of land and buildings	-9100		Item #1081	-9100	
			Item #1082	0	
			Item #1091 – #1090	0	
Total non-recurring non-operating items		(-160400)			(-160400)
Less: tax and minority interests		(29300)			(29300)
Thomson non-recurring earnings		11800	Item #210		11800
Less additional management exclusion	ons				
Stock provisions and write offs	-42000				
Discontinued operations	-10100				
Aggregate management exclusions		(-52100)			
Less: additional tax and minority interests		(400)			
Non-GAAP earnings		63500			

References

Accounting Standards Board. (1992). Financial Reporting Standard No. 3: Reporting Financial Performance. London: ASB.

Barth, M., & Kallapur, S. (1996). The effects of cross-sectional scale differences on regression results in empirical accounting research. *Contemporary Accounting Research*, 13(2), 527–567.

- Bhattacharya, N., Black, E., Christensen, T., & Larson, C. (2003). Assessing the relative informativeness and permanence of pro forma earnings and GAAP operating earnings. *Journal of Accounting and Economics*, 36(3), 285–319.
- Bradshaw, M., & Sloan, R. (2002). GAAP versus The Street: An empirical assessment of two alternative definitions of earnings. *Journal of Accounting Research*, 40(1), 41–66.
- Brown, L., & Sivakumar, K. (2003). Comparing the value relevance of two operating income measures. *Review of Accounting Studies*, 8(4), 561–572.



- Burgstahler, D., & Eames, M. (2003). Earnings management to avoid losses and earnings decreases: Are analysts fooled? *Contemporary Accounting Research*, 20(2), 253–294.
- Business Week. (2001). Time to cut the accounting shenanigans. May 14, 146.
- Collins, D., Maydew, E., & Weiss, I. (1997). Changes in the value-relevance of earnings and book values over the past forty years. *Journal of Accounting and Economics*, 24(1), 39–67.
- Davies, M., Paterson, R., & Wilson, A. (1999). U.K. GAAP: Generally accepted accounting practice in the United Kingdom (6th ed.). Croydon: Butterworths-Tolley.
- Doyle, J., Lundholm, R., & Soliman, M. (2003). The predictive value of expenses excluded from pro forma earnings. *Review of Accounting Studies*, 8(2–3), 145–174.
- Doyle, J., & Soliman, M. (2004). Do managers define 'street' earnings to meet or beat analyst forecasts? Unpublished working paper, University of Utah and Stanford University.
- Gu, Z., & Chen, T. (2004). Analysts' treatment of nonrecurring items in street earnings. Journal of Accounting and Economics, 38, 129–170.
- Johnson, B., & Schwartz, W. (2005). Are investors misled by pro forma earnings? Contemporary Accounting Research, 22(4), 915–963.
- Landsman, W., Miller, B., & Yeh, S. (2006). Implications of components of income excluded from pro forma earnings for future profitability and equity valuation. Unpublished working paper, University of North Carolina at Chapel Hill, University of California – Los Angeles, and National Taiwan University.
- Lin, S., & Walker, M. (2000). FRS3 earnings, headline earnings, and accounting-based valuation models. Accounting and Business Research, 30(4), 299–306.
- Lougee, B., & Marquardt, C. (2004). Earnings informativeness and strategic disclosure: An empirical examination of 'pro forma' earnings. *The Accounting Review*, 79(3), 769–795.
- Lipe, R. (1986) The information contained in the components of earnings. Journal of Accounting Research, 24(3), 37–64.
- Marques, A. (2006). SEC interventions and the frequency and usefulness of non-GAAP financial measures. *Review of Accounting Studies*, 11(4), 549–574.
- Matsumoto, D. (2002). Management's incentives to avoid earnings surprises. *The Accounting Review*, 77(3), 483–514.
- McVay, S. (2006). Earnings management using classification shifting: An examination of core earnings and special items. *The Accounting Review*, 81(3), 501–531.
- Newbold, P. (1988). Statistics for business and economics (2nd ed.). Prentice-Hall.
- Smith, T. (1992). Accounting for growth. London: Century Business.

_i\

2 Springer

- Turner, L. (2000). Remarks to the 39th Annual Corporate Counsel Institute. Presented at Northwestern University School of Law, October 12, 2000.
- Vuong, Q. (1989). Likelihood ratio tests for model selection and non-nested hypotheses. *Econometrica*, 57(2), 307–333.
- Walker, M., & Louvari, E. (2003). The determinants of voluntary disclosures of adjusted earnings per share measures by UK quoted companies. Accounting and Business Research, 33(4), 295–309.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

